Author's response to reviews

Title: Pain, power and patience - a narrative study of general practitioners' relations with chronic pain patients

Authors:

Mia Hemborg Kristiansson (Mia.Hemborg_Kristiansson@med.lu.se)
Annika Brorsson (Annika.Brorsson@med.lu.se)
Caroline Wachtler (carolinewachtler@hotmail.com)
Margareta Troein (Margareta.Troein@med.lu.se)

Version: 4 Date: 1 May 2011

Author's response to reviews:

We have made all the changes you required.

Best regards
Margareta Troein